% file: bibtest.tex     
% just some text and a lot of meaningless citations for testing purposes
\texsis

% page layout
\nopagenumbers
\Afour

% It's not the original BibTeX logo, but ... :-)
\def\BibTeX{BIB\TeX}

% The text: 
%
\nosechead{Testing \BibTeX{} with \TeXsis{}}

% These are NEW:
% specify the list of .bib files (without the .bib)
\ReferenceFiles{unix,internet,mylist}

% specify the .bst file (guess without what?)
\ReferenceStyle{texsis}


% to change the reference list layout
\def\vol#1{\bf #1}
%\Eurostyletrue

% for the citation marks in the text
%\superrefsfalse                % [n] reference marks


This is an example\cite{Droms:SPE-20-S2-119} text for testing the use
of \BibTeX{}\cite{Angebranndt:SPE-20-S2-69} in conjunction with the
\TeXsis{} macro package.  The citations\cite{Gajewska:SPE-20-S2-137} in
this paragraph\cite{Ross:IEEE-Comm-Mag-24-5-10} are just for fun,
without any special meaning\cite{Johnson:tools87}\cite{Droms:SPE-20-S2-119}.

Another stupid\cite{Dammann:EuroPhysLett21}\cite{Dammann:ZPhysB98}
citation.

% print the references
%
\nosechead{References} 
\ListReferences

% Say goodbye, ...
\bye
